ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Engle, Robert F.; Manganelli, Simone

Working Paper Value at risk models in finance

ECB Working Paper, No. 75

Provided in Cooperation with: European Central Bank (ECB)

Suggested Citation: Engle, Robert F.; Manganelli, Simone (2001) : Value at risk models in finance, ECB Working Paper, No. 75, European Central Bank (ECB), Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/152509

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

EUROPEAN CENTRAL BANK

WORKING PAPER SERIES



WORKING PAPER NO. 75

VALUE AT RISK MODELS

BY SIMONE MANGANELLI AND ROBERT F. ENGLE

August 2001

EUROPEAN CENTRAL BANK

WORKING PAPER SERIES



WORKING PAPER NO. 75

VALUE AT RISK MODELS IN FINANCE

BY SIMONE MANGANELLI¹ AND ROBERT F. ENGLE²

August 2001

© European Central Bank, 2001

Address	Kaiserstrasse 29
	D-60311 Frankfurt am Main
	Germany
Postal address	Postfach 16 03 19
	D-60066 Frankfurt am Main
	Germany
Telephone	+49 69 1344 0
Internet	http://www.ecb.int
Fax	+49 69 1344 6000
Telex	411 144 ecb d

All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged. The views expressed in this paper are those of the authors and do not necessarily reflect those of the European Central Bank.

ISSN 1561-0810

Contents

Ab	ostract	4
No	on-technical summary	5
I	Introduction	6
2	VaR Methodologies 2.1 Parametric Models 2.2 Nonparametric Methods 2.3 Semiparametric Models	7 8 10 12
3	Expected Shortfall	20
4	Monte Carlo Simulation 4.1 Sumulation Study of the Threshold Choice for EVT 4.2 Comparison of quantile methods performance	22 22 24
5	Conclusion	28
Re	ferences	29
Ар	opendix A: Tables	31
Ар	opendix B: Figures	35
Eu	ropean Central Bank Working Paper Series	36

Abstract

Value at Risk (VaR) has become the standard measure that financial analysts use to quantify market risk. VaR is defined as the maximum potential change in value of a portfolio of financial instruments with a given probability over a certain horizon. VaR measures can have many applications, such as in risk management, to evaluate the performance of risk takers and for regulatory requirements, and hence it is very important to develop methodologies that provide accurate estimates.

The main objective of this paper is to survey and evaluate the performance of the most popular univariate VaR methodologies, paying particular attention to their underlying assumptions and to their logical flaws. In the process, we show that the Historical Simulation method and its variants can be considered as special cases of the CAViaR framework developed by Engle and Manganelli (1999). We also provide two original methodological contributions. The first one introduces the extreme value theory into the CAViaR model. The second one concerns the estimation of the expected shortfall (the expected loss, given that the return exceeded the VaR) using a simple regression technique.

The performance of the models surveyed in the paper is evaluated using a Monte Carlo simulation. We generate data using GARCH processes with different distributions and compare the estimated quantiles to the true ones. The results show that CAViaR models are the best performers with heavy-tailed DGP.

JEL Classification Codes: C22, G22

NON TECHNICAL SUMMARY

The increased volatility of financial markets during the last decade has induced researchers, practitioners and regulators to design and develop more sophisticated risk management tools. Value at Risk (VaR) has become the standard measure that financial analysts use to quantify market risk. VaR is defined as the maximum potential loss in value of a portfolio due to adverse market movements, for a given probability. The great popularity that this instrument has achieved is essentially due to its conceptual simplicity: VaR reduces the (market) risk associated with any portfolio to just one number, the loss associated to a given probability.

VaR measures can have many applications, and is used both for risk management and for regulatory purposes. In particular, the Basel Committee on Banking Supervision (1996) at the Bank for International Settlements imposes to financial institutions such as banks and investment firms to meet capital requirements based on VaR estimates. Providing accurate estimates is of crucial importance. If the underlying risk is not properly estimated, this may lead to a sub-optimal capital allocation with consequences on the profitability or the financial stability of the institutions.

The main objective of this paper is to survey the recent developments in VaR modelling. The focus of the paper is on the underlying assumptions and the logical flaws of the available methodologies and not on their empirical application. In order to facilitate comparison, we restrict our attention to univariate methodologies.

There are also two original theoretical contributions in this paper. The first one introduces the extreme value theory into the Conditional Autoregressive Value at Risk or CAViaR model introduced by Engle and Manganelli (1999). The second one concerns the estimation of the expected shortfall (the expected loss, given that the return exceeded the VaR) using a simple regression technique. In the survey, we show also how the Historical Simulation method is just a special case of the CAViaR models.

The performance of the models surveyed in the paper is evaluated using a Monte Carlo simulation. We generate data using GARCH processes with different distributions and compare the estimated quantiles to the true ones. The results show that CAViaR models are the best performers with heavy-tailed DGP.

1. Introduction

Financial institutions are subject to many sources of risk. Risk can be broadly defined as the degree of uncertainty about future net returns. A common classification reflects the fundamental sources of this uncertainty. Accordingly, the literature distinguishes four main types of risk. *Credit risk* relates to the potential loss due to the inability of a counterpart to meet its obligations. It has three basic components: credit exposure, probability of default and loss in the event of default. *Operational risk* takes into account the errors that can be made in instructing payments or settling transactions, and includes the risk of fraud and regulatory risks. *Liquidity risk* is caused by an unexpected large and stressful negative cash flow over a short period. If a firm has highly illiquid assets and suddenly needs some liquidity, it may be compelled to sell some of its assets at a discount. *Market risk* estimates the uncertainty of future earnings, due to the changes in market conditions.

The most prominent of these risks in trading is market risk, since it reflects the potential economic loss caused by the decrease in the market value of a portfolio. Value at Risk (VaR) has become the standard measure that financial analysts use to quantify this risk. It is defined as the maximum potential loss in value of a portfolio of financial instruments with a given probability over a certain horizon. In simpler words, it is a number that indicates how much a financial institution can lose with probability θ over a given time horizon. The great popularity that this instrument has achieved among financial practitioners is essentially due to its conceptual simplicity: VaR reduces the (market) risk associated with any portfolio to just one number, that is the loss associated with a given probability.

VaR measures can have many applications, such as in risk management, to evaluate the performance of risk takers and for regulatory requirements. In particular, the Basel Committee on Banking Supervision (1996) at the Bank for International Settlements imposes to financial institutions such as banks and investment firms to meet capital requirements based on VaR estimates. Providing accurate estimates is of crucial importance. If the underlying risk is not properly estimated, this may lead to a sub-optimal capital allocation with consequences on the profitability or the financial stability of the institutions.

From a statistical point of view, VaR estimation entails the estimation of a quantile of the distribution of returns. The fact that return distributions are not constant over time poses exceptional challenges in the estimation. The main objective of this paper is to survey the recent developments in VaR modelling. We want to stress that the focus of the paper is on the underlying assumptions and the logical flaws of the available methodologies and not on their empirical application. In order to facilitate comparison, we restrict our attention to univariate methodologies. There are two original contributions in this paper. The first one introduces the extreme value theory into the Conditional Autoregressive Value at Risk or CAViaR model introduced by Engle and Manganelli (1999). The second one concerns the estimation of the expected shortfall (the expected loss, given that the return exceeded the VaR) using a simple regression technique. In the survey, we show also how the Historical Simulation method is just a special case of the CAViaR models.

The performance of the models surveyed in the paper is evaluated using a Monte Carlo simulation. We generate data using GARCH processes with different distributions and compare the estimated quantiles to the true ones. The results show that CAViaR models are the best performers with heavy-tailed DGP.

The rest of the paper is organised as follows. Section 2 contains the survey. Section 3 discusses several procedures to estimate the expected shortfall. Section 4 describes the implementation and the findings of the Monte Carlo simulation and section 5 briefly concludes.

2. VaR Methodologies

While VaR is a very easy and intuitive concept, its measurement is a very challenging statistical problem. Although the existing models for calculating VaR employ different methodologies, they all follow a common general structure, which can be summarised in three points: 1) Mark-to-market the portfolio, 2) Estimate the distribution of portfolio returns, 3) Compute the VaR of the portfolio.

The main differences among VaR methods are related to point 2, that is the way they address the problem of how to estimate the possible changes in the value of the portfolio. CAViaR models skip the estimation of the distribution issue, as they allow computing directly the quantile of the distribution. We will classify the existing models into three broad categories:¹

- Parametric (RiskMetrics and GARCH)
- Nonparametric (Historical Simulation and the Hybrid model)
- Semiparametric (Extreme Value Theory, CAViaR and quasi-maximum likelihood GARCH)

¹ The number and types of approaches to VaR estimation is growing exponentially and it's impossible to take all of them into account. In particular, Monte Carlo simulation and stress testing are commonly used methods that won't be discussed here. We refer the interested reader to the excellent web site *www.gloriamundi.org* for a comprehensive listing of VaR contributions.

The results that each of these methods yield can be very different from each other. Beder (1995) applies eight common VaR methodologies² to three hypothetical portfolios. The results show that the differences among these methods can be very large, with VaR estimates varying by more that 14 times for the same portfolio. Hence, in order to decide which methodology to choose, it is necessary to understand the underlying assumptions, as well as the mathematical models and quantitative techniques used. Only after this preliminary step, can the researcher choose the model that she thinks to be closer to her beliefs or objectives.

The basic inspiration of the VaR methodologies to be discussed below usually comes from the characteristics of financial data. The empirical facts about financial markets are very well known, since the pioneering works of Mandelbrot (1963) and Fama (1965). They can be summarised as follows:

1. Financial return distributions are leptokurtotic, that is they have heavier tails and a higher peak than a normal distribution.

2. Equity returns are typically negatively skewed.

3. Squared returns have significant autocorrelation, i.e. volatilities of market factors tend to cluster. This is a very important characteristic of financial returns, since it allows the researcher to consider market volatilities as quasistable, changing in the long run, but stable in the short period. Most of the VaR models make use of this quasi-stability to evaluate market risk.

All of the traditional VaR models try to account for some or all of these empirical regularities.

Throughout the rest of the paper we denote the time series of portfolio returns by $\{y_t\}_{t=1}^T$ and the θ -quantile of the return distribution at time *t* by $q_{t\theta}$. Estimated quantities are denoted by a hat (^).

2.1. Parametric Models

Models such as RiskMetrics (1996) and GARCH propose a specific parameterisation for the behaviour of prices. The family of ARCH models was introduced by Engle (1982) and Bollerslev (1986) and has been successfully applied to financial data. See Bollerslev, Engle and Nelson (1994) for a survey. The simplest GARCH (1,1) can be described as follows:

² EVT and CAViaR are not considered by Beder (1995) as they have been applied to VaR estimation only recently.

(1)
$$y_t = \sigma_t \varepsilon_t \qquad \varepsilon_t \sim i.i.d.(0,1)$$
$$\sigma_t^2 = \omega + \alpha y_{t-1}^2 + \beta \sigma_{t-1}^2$$

This model has two crucial elements: the particular specification of the variance equation and the assumption that the standardised residuals are i.i.d. The first element was inspired by the characteristics of financial data discussed above. The assumption of i.i.d. standardised residuals, instead, is just a necessary device to estimate the unknown parameters. A further necessary step to implement any GARCH algorithm is the specification of the distribution of the ϵ_i . The most generally used distribution is the standard normal³. Only after this extra distributional assumption has been imposed, does it become possible to write down a likelihood function and get an estimate of the unknown parameters. Once the time series of estimated variance is computed, the 5% quantile, say, is simply computed as -1.645 (the 5% quantile of the standard normal) times the estimated standard deviation.

Under the RiskMetrics approach the variance is computed using an Exponentially Weighted Moving Average, which correspond to an Integrated GARCH model: $\sigma_t^2 = \lambda \sigma_{t-1}^2 + (1-\lambda)y_{t-1}^2$, with λ usually set equal to 0.94 or 0.97. RiskMetrics also assumes that standardised residuals are normally distributed (see the Technical Document 1996 for details).

The general finding is that these approaches (both normal GARCH and RiskMetrics) tend to underestimate the Value at Risk, because the normality assumption of the standardised residuals seems not to be consistent with the behaviour of financial returns. The main advantage of these methods is that they allow a complete characterisation of the distribution of returns and there may be space for improving their performance by avoiding the normality assumption. On the other hand, both GARCH and RiskMetrics are subject to three different sources of misspecification: the specification of the variance equation and the distribution chosen to build the log-likelihood may be wrong, and the standardised residuals may not be i.i.d. Whether or not these misspecification issues are relevant for VaR estimation purposes is mainly an empirical issue.

³ This assumption is also motivated by the very important result of Bollerslev and Woolridge (1992) on quasi-maximum likelihood GARCH. More on this later.

2.2. Nonparametric Methods

One of the most common methods for VaR estimation is the Historical Simulation. This approach drastically simplifies the procedure for computing the Value at Risk, since it doesn't make any distributional assumption about portfolio returns. Historical Simulation is based on the concept of rolling windows. First, one needs to choose a window of observations, that generally ranges from 6 months to two years. Then, portfolio returns within this window are sorted in ascending order and the θ -quantile of interest is given by the return that leaves θ % of the observations on its left side and $(1-\theta)$ % on its right side. If such a number falls between two consecutive returns, then some interpolation rule is applied. To compute the VaR the following day, the whole window is moved forward by one observation and the entire procedure is repeated.

Even if this approach makes no explicit assumptions on the distribution of portfolio returns, an implicit assumption is hidden behind this procedure: the distribution of portfolio returns doesn't change within the window. From this implicit assumption several problems derive.

First, this method is logically inconsistent. If all the returns within the window are assumed to have the same distribution, then the logical consequence must be that all the returns of the time series must have the same distribution: if $y_{t-window}$..., y_t and $y_{t+1-window}$..., y_{t+1} are i.i.d., then also y_{t+1} and $y_{t-window}$ must be i.i.d., by the transitive property. Second, the empirical quantile estimator is consistent only if k, the window size, goes to infinity. The third problem concerns the length of the window. This is a very delicate issue, since forecasts of VaR under this approach are meaningful only if the historical data used in the calculations have (roughly) the same distribution. In practice, the volatility clustering period is not easy to identify. The length of the window must satisfy two contradictory properties: it must be large enough, in order to make statistical inference significant, and it must not be too large, to avoid the risk of taking observations outside of the current volatility cluster. Clearly, there is no easy solution to this problem.

Moreover, assume that the market is moving from a period of relatively low volatility to a period of relatively high volatility (or vice versa). In this case, VaR estimates based on the historical simulation methodology will be biased downwards (correspondingly upwards), since it will take some time before the observations from the low volatility period leave the window.

Finally, VaR estimates based on historical simulation may present predictable jumps, due to the discreteness of extreme returns. To see why, assume that we are computing the VaR of a portfolio using a rolling window of 180 days and that today's return is a large negative number. It is easy to predict that the VaR estimate will jump upward, because

of today's observation. The same effect (reversed) will reappear after 180 days, when the large observation will drop out of the window. This is a very undesirable characteristic and it is probably enough to discard the historical simulation method as a reliable one.

An interesting variation of the historical simulation method is the hybrid approach proposed by Boudoukh, Richardson and Whitelaw (1998). The hybrid approach combines RiskMetrics and historical simulation methodologies, by applying exponentially declining weights to past returns of the portfolio. There are three steps to follow. First, to each of the most recent *K* returns of the portfolio, y_b y_{t-1} , ..., y_{t-K+1} , is associated a weight, $\frac{1-\lambda}{1-\lambda^K}, \left(\frac{1-\lambda}{1-\lambda^K}\right)\lambda, ..., \left(\frac{1-\lambda}{1-\lambda^K}\right)\lambda^{K-1}$, respectively⁴. Second, the returns are ordered in ascending order. Third, to find the $\theta\%$ VaR of the portfolio, sum the corresponding weights until $\theta\%$ is reached, starting from the lowest return. The VaR of

the portfolio is the return corresponding to the last weight used in the previous sum⁵.

We believe that this approach constitutes a significant improvement over the methodologies discussed so far, since it drastically simplifies the assumptions needed in the parametric models and it incorporates a more flexible specification than the historical simulation approach. To better understand the assumptions behind this method, note that it implies the following:

(2)
$$\hat{q}_{t+1,\theta} = \sum_{j=t-K+1}^{t} y_j I \left(\sum_{i=1}^{K} f_i(\lambda; K) I(y_{t+1-i} \le y_j) = \theta \right)$$

where $f_i(\lambda;K)$ are the weights associated with return y_i and $I(\bullet)$ is the indicator function. If we set $f_i(\lambda;K)=1/K$ we get as a special case the Historical Simulation estimator.⁶ The main difference between these methods is in the specification of the quantile process. In the Historical Simulation method, each return is given the same weight, while in the Hybrid method different returns have different weights, depending on how old the observations are.

Hence, strictly speaking, none of these models is nonparametric, as a parametric specification is proposed for the quantile. This is the same idea behind CAViaR methodology. Boudoukh, Richardson and Whitelaw set λ equal to 0.97

⁴ Note that the role of the term $\frac{1-\lambda}{1-\lambda^{K}}$ is simply to ensure that the weights sum to 1.

⁵ To achieve exactly the θ % of the distribution, adjacent points are linearly interpolated.

⁶ Here, for simplicity of notation but without loss of generality, we assume that no interpolation rule is necessary.

and 0.99, as in their framework no statistical method is available to estimate this unknown parameter. Of course, by recognising that both the Hybrid and Historical Simulation models are just a special case of CAViaR models, one can use the regression quantile framework to estimate the parameter λ and the window length K.⁷

2.3. Semiparametric Models

Recently, alternative methods have been proposed to estimate Value at Risk, such as applications of Extreme Value Theory (see, for example, Danielsson and deVries (1998) or Gourieroux and Jasak (1998)) and applications of regression quantile technique such as in Chernozhukov and Umantsev (2000) and Engle and Manganelli (1999). Other approaches that can be included under this section are those based on Bollerslev and Woolridge (1992) quasi-maximum likelihood GARCH, as suggested by Diebold, Schuermann and Stroughair (1999), and independently implemented by McNeil and Frey (2000) and Engle and Manganelli (1999)⁸. A similar idea can be found in Hull and White (1998), although their model is not as robust as those cited above, since the variance is not estimated using QML GARCH.

a) Extreme Value Theory

EVT is a classical topic in probability theory. Many books and surveys are available on the subject, see for example Leadbetter, Lindgren and Rootzen (1983) or Embrechts, Kluppelberg and Mikosch (1997). Here we intend to give some intuition and basic results of EVT, following very closely the approach of Embrechts, Kluppelberg and Mikosch (1997). EVT can be conveniently thought as a complement to the Central Limit Theory: while the latter deals with fluctuations of cumulative sums, the former deals with fluctuations of sample maxima. The main result is due to Fisher and Tippett (1928), who specify the form of the limit distribution for appropriately normalised maxima. Let $M_n = max\{X_1, ..., X_n\}$, where $\{X_i\}_{i=1}^n$ is a sequence of i.i.d. random variables. Fisher and Tippett showed that if there exist norming constants

 $c_n > 0, d_n \in \mathcal{R}$ and some non-degenerate distribution function H such that $c_n^{-1}(M_n - d_n) \xrightarrow{a} H$, then

(3)
$$H = H_{\xi}(x) = \exp\left\{-\left(1 + \xi x\right)^{-\frac{1}{\xi}}\right\} \qquad 1 + \xi x > 0$$

⁷ Although there might be some theoretical issues, due to the discontinuity of the indicator function.

⁸ See the approach proposed at the end of section 9.

The special case $H_0(x)$ is to be interpreted as $\lim_{\xi \to 0} H_{\xi}(x)$. *H* is called the Generalised Extreme Value (GEV) distribution and describes the limit distribution of normalised maxima. An important concept for the application of EVT to VaR estimation is Maximum Domain of Attraction: the random variable *X* belongs to the Maximum Domain of Attraction of the extreme value distribution *H* (and we write $X \in MDA(H)$) if and only if Fisher-Tippett theorem holds for *X*, with limit distribution *H*.

 ξ is a crucial parameter that determines the shape of the GEV distribution. Distributions that belong to $MDA(H_{\xi})$, for $\xi>0$, are called heavy tailed (examples are Cauchy, Student-t, Pareto, loggamma). Gamma, normal, lognormal and exponential distributions belong to $MDA(H_0)$, while distributions with finite endpoints (such as the uniform and beta) belong to $MDA(H_{\xi})$, for $\xi<0$.

There are two alternative ways to implement EVT. The first one is generally built around the Hill estimator. We won't discuss it here and we refer the interested reader to Embrechts, Kluppelberg and Mikosch (1997) and to Danielsson and de Vries (1997).

The second one is based on the concept of exceptions of high thresholds. In this paper we adopt this estimation method because the Hill estimator is designed only for data coming from heavy tailed distributions⁹, while in the Monte Carlo simulation of section 2.4.1 we consider also normal and gamma distributions. Moreover, estimates based on this second approach tend to be more stable than the Hill estimator with respect to the choice of the threshold, as shown by McNeil and Frey (2000).

Suppose X_i are i.i.d. random variables with distribution function $F \in MDA(H_{\xi})$. Define the excess distribution function as $F_u(y) = P(X_i - u \le y | X_i > u)$ for $y \ge 0$ and for a given threshold u. Define also the Generalised Pareto Distribution (GPD) as:

(4)
$$G_{\xi,\beta}(x) = 1 - \left(1 + \xi x / \beta\right)^{-\frac{1}{\xi}} \qquad \beta > 0 \text{ and } \begin{cases} x \ge 0 & \text{if } \xi > 0\\ 0 \le x \le -\beta / \xi & \text{if } \xi < 0 \end{cases}$$

where again we use the convention $G_{0,\beta}(x) = \lim_{\xi \to 0} G_{\xi,\beta}(x)$. The following is a crucial property of GPD, due to Pickands (1975):

⁹ Although extensions that cover non fat-tailed distributions have been proposed; see, for example, Dekkers, Einmahl and de Haan (1989).

$$F \in MDA(H_{\xi}) \qquad \Leftrightarrow \qquad \lim_{u \uparrow x_F} \sup_{0 < x < x_F - u} \left| F_u(x) - G_{\xi,\beta}(x) \right| = 0$$

where x_F is the right endpoint (which could be infinite) of the distribution *F*. This result states that if *F* is in the Maximum Domain of Attraction of a GPD, as the threshold *u* approaches the endpoint of *F*, GPD becomes an accurate approximation of the excess distribution function F_u .

The parameters of the GPD can be estimated by maximum likelihood, once the threshold has been chosen. The log-likelihood based on (4) computed on the *N* upper order statistics is:

(5)
$$\max_{\xi,\beta} \left\{ -N \ln \beta - (1+1/\xi) \sum_{i=1}^{N} \ln \left(1 + \frac{\xi}{\beta} \left(X_{k_i} - u \right) \right) \right\} \\ \sum_{i=1}^{N} \ln \left\{ \left[1(\xi > 0) l \left(X_{k_i} - u > 0 \right) + 1(\xi < 0) l \left(0 \le X_{k_i} - u \le -\frac{\beta}{\xi} \right) \right] l(\beta > 0) \right\}$$

where N=N(u) denotes the number of observation exceeding the threshold u and $X_{k_1}, X_{k_2}, ..., X_{k_N}$ are the upper order statistics exceeding this threshold. Note that the second part of the log-likelihood simply reflects the domain constraint of the GPD. Noting that

1-
$$F(u+y) = [1-F(u)][1-F_u(y)]$$

≈[1- $F(u)][1-G_{ξβ}(y)]$

an estimator for the tail of 1-F(u+y) for y>0 can be easily derived. The term 1-F(u) can be estimated using the empirical distribution function: $[1-F(u)]^{n}=N/n$.

Substituting the maximum likelihood estimates of ξ and β in the GPD, we get:

(6)
$$[1 - F(u + y)]^{h} = \left(1 + \hat{\xi} \frac{\hat{y}}{\hat{\beta}}\right)^{-1/\hat{\xi}} N/n$$

The \hat{x}_p quantile estimator is now obtained by simply inverting the above formula:

(7)
$$\hat{x}_p = u + \left[\left((1-p)\frac{n}{N} \right)^{-\hat{\xi}} - 1 \right] \frac{\hat{\beta}}{\hat{\xi}}$$

Note that this estimator is valid only for very low *p*, as the approximation is valid only asymptotically.

To summarise, EVT seems to be a very general approach to tail estimation. The main strength is that the use of a GEV distribution to parameterise the tail doesn't seem to be a very restrictive assumption, as it covers most of the commonly used distributions. On the other hand, there are several problems that need to be considered.

First, the assumption of i.i.d. observations seems to be at odds with the characteristics of financial data. Although generalisations to dependent observations have been proposed (see, for example, Leadbetter, Lindgren, and Rootzen (1983) or Embrechts, Kluppelber and Mikosch (1997)), they either estimate the marginal unconditional distribution or impose conditions that rule out the volatility clustering behaviour typical of financial data. A solution to this problem is suggested in the following sections.

Second, EVT works only for very low probability levels. How low these probability levels must be is hard to tell on a priori ground. A Monte Carlo study might help to shed some light on how fast the performance of the EVT estimators deteriorates as we move away from the tail.

Closely related to this issue is the selection of the cut-off point that determines the number of order statistics to be used in the estimation procedure. The choice of the threshold u presents the same problems encountered in the choice of the number of k upper order statistics that enter the Hill estimator. If the threshold is too high, there are too few exceptions and the result is a high variance estimator. On the other hand, a threshold too low produces a biased estimator, because the asymptotic approximation might become very poor. Unfortunately, as of today there is no statistical method to choose u and one usually relies on simulations and graphical devices. Recently Danielsson and de Vries (1997) and Danielsson, de Haan, Peng and de Vries (1998) have proposed a two step bootstrap method to select the sample fraction on which the Hill estimator is based. Although their result is based on an assumption on the second order property of F that is rarely verifiable in practice, it is certainly an important contribution that can supplement the usual graphical analysis and that can stimulate further research on this very important issue. For example, it would be interesting to know whether their result can be extended also to the GPD approach.

Another issue is that the choice of k based on the minimisation of the mean squared error results in a biased estimator. As the quantile is a non-linear function of the tail estimator, it is important to quantify the size of this bias. Very little attention has been devoted to this problem. Since Value at Risk is the ultimate object of interest, and a bad estimation of it might have very relevant consequences on the profitability of a financial institution, we believe this to be a topic that deserves further exploration.

We will address some of these issues later on.

b) CAViaR

The Conditional Autoregressive Value at Risk, or CAViaR model was introduced by Engle and Manganelli (1999). The basic intuition is to model directly the evolution of the quantile over time, rather than the whole distribution of portfolio returns. They propose the following specification for the Value at Risk:

(8)
$$q_{t,\theta} = \beta_0 + \beta_1 q_{t-1,\theta} + l(\beta_2, ..., \beta_p, y_{t-1}, q_{t-1,\theta})$$

Different models can be estimated by choosing different specifications for the *l* function. Some of the models considered in the original paper are the Symmetric Absolute Value, $l(\cdot) = \beta_2 |y_{t-1}|$ and the Asymmetric Slope, $l(\cdot) = \beta_2 1(y_{t-1} > 0) - \beta_3 1(y_{t-1} < 0)$. A CAViaR model useful for Monte Carlo simulation is the Indirect GARCH(1,1):

(9)
$$q_{t,\theta} = \left(\beta_1 + \beta_2 q_{t-1,\theta}^2 + \beta_3 y_{t-1}^2\right)^{\frac{1}{2}}$$

A special case of CAViaR is the model suggested by Chernozhukov and Umantsev (2000), who set $\beta_1=0$ and $l(\cdot)=X_t'\beta$, where X_t is a vector containing observable economic variables at time t and β a vector of parameters to be estimated.

The unknown parameters are estimated using non-linear regression quantile techniques.¹⁰ Consider the following model:

¹⁰ Note that the quantile specification proposed by Chernozhukov and Umantsev (2000) is linear in the parameters. Under this assumption, the objective function (11) can be rewritten as a linear programming problem. This guarantees that the quantile regression estimate will be obtained in a finite number of simplex iterations. See, for example, Buchinsky (1998) for more details.

(10)
$$y_t = q_{t\theta} + \varepsilon_{t\theta}$$
 $Quant_{\theta}(\varepsilon_{t\theta}|\Omega_t) = 0$

where θ is the probability level associated with the quantile, Ω_t is the information set at time t and $Quant_{\theta}(\varepsilon_{t\theta}|\Omega_t)=0$ simply means that the θ -quantile of the error term is 0. White (1994) has shown that the minimisation of the regression quantile objective function introduced by Koenker and Bassett (1978) is able to deliver consistent estimates, under suitable assumptions:

(11)
$$\min_{\beta} \frac{1}{T} \left\{ \sum_{t: y_t \ge q_{t,\theta}(\beta)} \theta \mid y_t - q_{t,\theta}(\beta) \mid + \sum_{t: y_t < q_{t,\theta}(\beta)} (1-\theta) \mid y_t - q_{t,\theta}(\beta) \mid \right\}$$

Engle and Manganelli (1999) show also how to derive the asymptotic distribution of the estimator in order to perform hypothesis testing. We refer to the original paper for further details.

The only assumption required under this framework is that the quantile process is correctly specified. In particular, no assumption on the distribution of the error terms is needed, hence reducing the risk of misspecification. Moreover, even if the quantile process is misspecified, the minimisation of the regression quantile objective function can still be interpreted as the minimisation of the Kullback-Leibler Information Criterion, which measures the discrepancy between the true model and the one under study (for a treatment of quasi-maximum likelihood estimation, see White (1994)).

c) Quasi-Maximum Likelihood GARCH

In our discussion of GARCH models, we stressed how the assumption of normally distributed standardised residuals seemed to be at odds with the fact that financial data tend to exhibit heavy tails. It turns out, however, that the normality assumption might not be as restrictive as one can think. This is due to a very important result by Bollerslev and Woolridge (1992) who showed that the maximisation of the normal GARCH likelihood is able to deliver consistent estimates, provided that the variance equation is correctly specified, even if the standardised residuals are not normally distributed. We refer to this result as the quasi-maximum likelihood (QML) GARCH. Many papers have exploited this

property. Engle and Gonzalez-Rivera (1991), for example, propose a semiparametric estimation of GARCH models. Engle and Manganelli (1999), at the end of section 9, suggest computing the VaR of a portfolio by first fitting a QML GARCH and then multiplying the empirical quantile of the standardised residuals by the square root of the estimated variance. This estimation method is a mix of a GARCH fitted to portfolio returns and historical simulation applied to the standardised residuals. As a consequence it retains some of the drawbacks of these approaches. First, the assumption that the standardised residuals are i.i.d. is still required. Given this assumption, however, the logical inconsistency of historical simulation is solved, as they use as a window the whole series of standardised residuals. On the other hand, the problem of discreteness of extreme returns remains. Historical simulation will provide very poor and volatile quantile estimates in the tails, for the same reason that kernel methods don't guarantee to provide an adequate description of the tails. A natural alternative seems to use EVT instead of historical simulation. This model was estimated by McNeil and Frey (2000).

Given the very general results of EVT, the QML GARCH augmented by the EVT requires very weak assumptions. The only required assumptions are that the variance is correctly specified and that the standardised residuals are i.i.d. and in the maximum domain of attraction of some extreme value distribution.

d) CAViaR and Extreme Value Theory

Even if more empirical work needs to be done, there is preliminary evidence that CAViaR models work fine at very common probability levels, such as 5% and 1%. However, there might be some problems (essentially due to finite sample size) when one tries to estimate very extreme quantiles, such as 0.1% or even lower. This has been documented in a Monte Carlo study by Engle and Manganelli (1999). A viable alternative could be to incorporate EVT into the CAViaR framework.

The strategy we adopt is the following. We fit a CAViaR model to get an estimate of the quantile $q_{t,\theta}$ with θ large enough (between 5% and 10%, say) so that sensible estimates can be obtained. Then we construct the series of standardised quantile residuals:

(12)
$$\hat{\varepsilon}_{t,\theta} / \hat{q}_{t,\theta} = (y_t / \hat{q}_{t,\theta}) - 1$$

Let $g_t(\cdot)$ be the conditional density function of the standardised quantile residuals. We impose the following assumption.

A1 -
$$g_t \begin{pmatrix} \varepsilon_{t,\theta} \\ q_{t,\theta} \end{pmatrix} = g \begin{pmatrix} \varepsilon_{t,\theta} \\ q_{t,\theta} \end{pmatrix}$$
, for all $\frac{\varepsilon_{t,\theta}}{q_{t,\theta}} > 0$, for all *t*, provided that $q_{t,\theta} \neq 0$ for all *t*.

That is, the distribution of the standardised quantile residuals is not time-varying beyond the $(1-\theta)$ -quantile.¹¹

Under this assumption, we can now apply EVT to the standardised quantile residuals to get an estimate of the tail.

(13)
$$\Pr\left(y_t < q_{t,p}\right) = \Pr\left(y_t < q_{t,\theta} - \left(q_{t,\theta} - q_{t,p}\right)\right) = \Pr\left(\frac{y_t}{q_{t,\theta}} - 1 > z\right) = p, \quad \text{for } p < \theta$$

where $z = \frac{q_{t,p}}{q_{t,\theta}} - 1$ is the (1-*p*)-quantile of the standardised quantile residuals. Assume that z_p is the EVT estimate of the (1-*p*)-quantile of the standardised residuals. The corresponding *p*-quantile of portfolio returns is then given by

 $\hat{q}_{t,p} = \hat{q}_{t,\theta} (1 + \hat{z}_p)$. Note that if $p = \theta$, then $z_p = 0$.

Another nice consequence of this assumption is the possibility to estimate the density function $h_t(\varepsilon_{t,\theta})$ that enters the variance-covariance matrix of regression quantile.¹² Hence, more efficient estimates of the parameters of the CAViaR model can be obtained by weighting the objective function by this density function evaluated at 0:

(14)
$$\min_{\beta} \frac{1}{T} \left\{ \sum_{t: y_t \ge q_{t,\theta}(\beta)} \theta \mid y_t - q_{t,\theta}(\beta) \mid h_t(0) + \sum_{t: y_t < q_{t,\theta}(\beta)} (1-\theta) \mid y_t - q_{t,\theta}(\beta) \mid h_t(0) \right\}$$

¹¹ Here we implicitly assume that $q_{t,\theta}$ is a negative number.

¹² See Engle and Manganelli (1999) for details.

Newey and Powell (1991) showed that by minimising this objective function, the estimator attains the semiparametric efficiency bound, when the quantile function is linear. It is likely that this result extends also to the non-linear case, but no such result exists yet.

Since assumption A1 imposes some restrictions on the error term of CAViaR models, we would like to stress the difference between the set of assumptions required by CAViaR and QML GARCH models. Assumption A1 requires that the standardised quantile residuals are i.i.d. only in the right tail, allowing for any sort of behaviour in the rest of the distribution. QML GARCH, instead, requires that the tails have the same dynamic behaviour as the rest of the distribution. Again, whether the extra restrictions imposed by QML GARCH models are relevant or not, is exclusively an empirical question.

3. Expected Shortfall

Many authors have criticised the adequacy of VaR as a measure of risk. Artzner et al. (1997, 1999) propose, as an alternative measure of risk, the Expected Shortfall, which measures the expected value of portfolio returns given that some threshold (usually the Value at Risk) has been exceeded. Analytically:

(15)
$$E_t \left(y_t \mid y_t < q_{t,\theta} \right)$$

Since σ_t is a function of variables known at time *t*, it is possible to rewrite (15) in terms of standardised residuals:

(16)
$$E_t \left(y_t \mid y_t < q_{t,\theta} \right) = \sigma_t E_t \left(\frac{y_t}{\sigma_t} \middle| \frac{y_t}{\sigma_t} < \frac{q_{t,\theta}}{\sigma_t} \right)$$

If we assume that the standardised residuals follow an EVT distribution, EVT provides a very simple formula to compute this value. In general, if *X* has a Generalised Pareto distribution with parameters $\xi < 1$ and β , then for $u < x_F$:

(17)
$$E(X-u \mid X > u) = \frac{\beta + \xi u}{1-\xi} \beta + u\xi > 0$$

It can be easily shown (see, for example, McNeil and Frey (2000) eq. 16) that for any $x_p > u$,

(18)
$$E(X \mid X > x_p) = x_p \left(\frac{1}{1-\xi} + \frac{\beta - \xi u}{(1-\xi)x_p}\right)$$

Suppose we fit a QML GARCH augmented with EVT and we get estimates for q_t and σ_t in (16). Then, applying the formula in (18) to compute the expectation in (16), we get:

(19)
$$\hat{E}_t\left(y_t \mid y_t < q_{t,\theta}\right) = \hat{q}_{t,\theta}\left(\frac{1}{1-\hat{\xi}} - \frac{\hat{\beta} - \hat{\xi}\hat{u}}{1-\hat{\xi}}\frac{\hat{\sigma}_t}{\hat{q}_{t,\theta}}\right)$$

A similar result can be obtained for CAViaR models. Assuming that $q_{t,\theta}$ is a negative number, from (15) we have:

(20)
$$E_t\left(y_t \mid y_t < q_{t,\theta}\right) = q_{t,\theta} E_t\left(\frac{y_t}{q_{t,\theta}} \mid \frac{y_t}{q_{t,\theta}} > 1\right)$$

Given the assumption that the standardised quantile residuals are i.i.d. beyond a certain threshold, we can again apply EVT to compute the expected shortfall:

(21)
$$\hat{E}_{t}\left(y_{t} \mid y_{t} < q_{t,\theta}\right) = \hat{q}_{t,\theta}\left(\frac{1}{1-\hat{\xi}} + \frac{\hat{\beta}-\hat{\xi}\,\hat{u}}{1-\hat{\xi}}\right)$$

Formulae (16) and (20) suggest an alternative way to compute the expected shortfall. Noting that the conditional expectation in both formulae must be constant (given the i.i.d. assumption), one can compute the expected shortfall by simply regressing the returns that exceed the quantile against the corresponding estimated standard deviation (or quantile):

(22)	$y_t = \beta \sigma_t + \eta_t$	for	$y_t < q_t$
------	---------------------------------	-----	-------------

(23) $y_t = \delta q_t + \eta_t$ for $y_t < q_t$

The expected shortfall is now given by, for QML GARCH and CAViaR models respectively:

(24)
$$\hat{E}_t(y_t \mid y_t < q_{t,\theta}) = \hat{\beta} \,\hat{\sigma}_t$$

(25)
$$\hat{E}_t \left(y_t \mid y_t < q_{t,\theta} \right) = \hat{\delta} \, \hat{q}_t$$

where $\hat{\beta}$ and $\hat{\delta}$ are the regression coefficients of (22) and (23).

We view this second estimation method as complementary to the EVT method. Indeed, this approach can be useful to compute the expected shortfall at relatively high quantiles (such as 5%) where the approximation provided by the EVT may be very poor. On the other hand, the regression method can give very poor estimates at very low quantiles, since only few observations are available. In the extreme case, when the quantile is beyond the boundary of observed data, no observation would be available, and hence the method wouldn't be applicable.

4. Monte Carlo Simulation

4.1. Simulation Study of the Threshold Choice for EVT

As we have already pointed out, a critical aspect for the implementation of EVT is the determination of the threshold beyond which the observations are assumed to follow a Generalised Pareto distribution. To address this issue we have performed a Monte Carlo simulation.

We generated 1000 samples of 2000 observations each, using three different distributions, the normal and the Student-t with 3 and 4 degrees of freedom. For each sample, we estimated the 5%, 1% and 0.05% quantiles using the formula of the extreme value theory provided above and using different threshold values. We chose the threshold values indirectly, by choosing the number of exceptions (k) to be included in the maximum likelihood estimation. We started with k=10 and we increased it by 5 until it reached 700. To compare the different estimates, we computed the bias and the mean squared error of the estimators as follows:

(26)
$$bias_{\theta,k} = \sum_{j=1}^{1000} \hat{q}_{\theta,k}^{(j)} / 1000 - q_{\theta}$$

(27)
$$MSE_{\theta,k} = \sum_{j=1}^{1000} \left(\hat{q}_{\theta,k}^{(j)} - q_{\theta} \right)^2 / 1000$$

Our goal is to determine how sensitive these estimates are to the choice of the parameter k, to the underlying distribution and to the confidence level of the estimated quantile. The results are reported in Figure 1. We plot the bias and MSE of the EVT estimates against k, together with the empirical quantile estimates, which don't depend on k.

A first result that emerges from these plots is that the 5% and 1% EVT estimates appear to be biased for very low k. This is probably due to the fact that these two confidence levels are not extreme enough for the GPD to be a reasonably good approximation of the tail of the underlying distribution. However, as k increases both the bias and the MSE of the EVT estimates are significantly reduced. In the normal case, the EVT MSE seems to be comparable with the empirical MSE for all values of k greater than 100 for the 5% quantile and for all values of k greater than 30 for the 1% quantile. The 5% and 1% EVT bias, instead, is close to the empirical bias for k between 100 and 350 and for k between 10 and 250, respectively, after which it grows rapidly in the negative direction. The plots look different for the Student-t distributions. In this case, the MSE of the 5% and 1% EVT rapidly deteriorates as k increases, while the bias remains more or less constant until k = 500. For these distributions a choice of k = 100 or even lower seems to be the most appropriate. Note also that the empirical quantile might be a better estimation method at these confidence levels as it has similar bias, a lower MSE and it does not depend on the choice of any parameter.

The situation changes as we move further out in the tail. For 0.05% quantile estimates, the empirical quantile becomes unreliable, as only few observations are available at these extreme confidence levels. In the Student-t case with 3 degrees of freedom, the empirical quantile has a greater bias than the EVT estimate and a MSE more than 10 times greater. The plots also show the bias-MSE trade-off for the EVT estimates. For the Student-t distributions, the bias is very close to 0 for low values of *k* and increases gradually as *k* increases. The MSE, instead, decreases until *k* equals about 400 and increases afterwards. It is hard to tell what is the optimal *k* under these circumstances. If one decides to minimise the MSE, a greater bias will result. On the other hand, by choosing a relatively low *k*, one obtains a more precise estimate on average, but at the cost of a greater variability. It would be helpful to be able to estimate the

bias that results from the minimisation of the MSE, but to our knowledge no such contribution is yet available in the literature.

4.2. Comparison of quantile methods performance

In this section, we compare the performance of the quantile models discussed in section 2 using a Monte Carlo simulation. We generated 1000 samples of 2000 observations for 7 different processes. The first 5 are GARCH processes with parameters [2.5, 0.04, 0.92]. The error terms were generated using random number generators with the following distributions: 1) standard normal, 2) Student-t with 3 degrees of freedom, 3) Student-t with 4 degrees of freedom, 4) Gamma with parameters (2,2), 5) Gamma with parameters (4,2). Note that in order to have error terms with mean zero and variance 1 (as required by GARCH models), the distributions above had to be standardised. For the Student-t distributions, we simply divided the random numbers by $\sqrt{v/(v-2)}$, the standard deviation of a Student-t with *v* degrees of freedom. For the Gamma (*a*,*b*) distributions, instead, we first subtracted the mean, *ab*, then divided by the standard deviation, $\sqrt{ab^2}$, and finally reversed the sign in order to have an infinite left tail.

We generated also two more processes. The first one is a process with a GARCH variance (with the same parameters as above) but with non-i.i.d. error terms that are alternatively drawn from a Student-t (3) and a Gamma (2,2). The second one is a CAViaR process, generated as follows. We start with an initial value for the quantile, q_0 , and generate the next θ -quantile as $q_{t,\theta} = -\sqrt{\omega + \alpha y_{t-1}^2 + \beta q_{t-1,\theta}^2}$, with parameters [2, 0.08, 0.9]. y_t is then generated from a distribution with θ -quantile equal to $q_{t,\theta}$. This is done by multiplying $q_{t,\theta}$ by a random number drawn from a distribution with (1- θ)-quantile equal to 1:

(28)
$$\theta = \Pr(y_t < q_{t,\theta}) = \Pr(q_{t,\theta}\varepsilon_t < q_{t,\theta}) = \Pr(\varepsilon_t > 1)$$

In the last term of (28), we reversed the sign of the inequality because $q_{t,\theta}$ is a negative number. The random number was drawn sequentially from a Student-t with 3 and 4 degrees of freedom and from a Gamma (2,2), and then divided by the θ -quantile of these distributions, to ensure that the (1- θ)-quantile was equal to 1. Note that the distributions were appropriately standardised so that they had mean zero and variance one, and that the θ -quantile of these standardised distributions is negative for low values of θ . Note also that this is not a GARCH process, and hence the parameters can be consistently estimated using a CAViaR model, but not with a GARCH model.

The performance of the models is evaluated using the equivalents of the bias and MSE for vector estimates:

(29)
$$bias_{\theta} = \left\| \sum_{j=1}^{1000} \left(\hat{q}_{\theta}^{(j)} - q_{\theta}^{(j)} \right) / 1000 \right\|^2 / 2000$$

(30)
$$MSE_{\theta} = \sum_{j=1}^{1000} \left(\left\| \hat{q}_{\theta}^{(j)} - q_{\theta}^{(j)} \right\|^2 / 2000 \right) / 1000$$

where $\|\cdot\|$ denotes the Euclidean norm, $q_{\theta}^{(j)}$ is the (2000×1) vector of true time-varying θ -quantiles and $\hat{q}_{\theta}^{(j)}$ is the (2000×1) vector of estimated θ -quantiles for the sample *j*.¹³ Note that for each sample we get 2000 quantile estimates and (30) measures the average distance of these estimates from the true quantile values.

We also tested the performance of the models in estimating the expected shortfall. From the true DGP, it is possible to compute the true expected shortfall¹⁴:

(31)
$$ES = E(y_t \mid y_t < VaR_t) = \theta^{-1} \int_{-\infty}^{VaR_t} y_t f_t(y_t) dy$$

where f_t is the pdf of y_t .

For each model, we estimated ES using EVT and/or the regression methods wherever they were applicable and computed the bias and MSE in the same way as shown in (29) and (30).¹⁵

We estimated the quantiles using 9 different methods:

1) *Historical simulation*, 2) *Normal GARCH* (the θ -quantile is computed as the estimated GARCH variance times the θ -quantile of the standard normal distribution), 3) *QML GARCH* (the θ -quantile is computed as the estimated

¹³ Other criteria might be used for the evaluation. For example, we could have used the L-1 norm instead of the L-2 norm, or an asymmetric criterion that gives higher weight to underestimated quantiles (based on the economic justification that the risk of underestimating the VaR is potentially more harmful than the cost of overestimating it). However, there is no clear a priori reason to prefer an L-1 norm to an L-2 norm, and economic intuition doesn't say anything about how to determine the weights in the asymmetric criterion.

¹⁴ The integral was evaluated using the command *quad8* in Matlab, which implements a quadrature algorithm using an adaptive recursive Newton-Cotes 8 panel rule. To compute the integral, we had to modify the pdf of the original Student-t and Gamma distributions using the Jacobian of the transformation, because the random variables were drawn from standardized distributions.

¹⁵ For the regression method, when no observation beyond VaR was available, we set ES equal to infinity and no value is reported in the table.

GARCH variance times the θ empirical quantile of the standardised residuals), 4) *QML GARCH & EVT* (the θ -quantile is computed as the estimated GARCH variance times the θ EVT quantile of the standardised residuals), 5) *EVT*, 6) *CAViaR*, 7) *Efficient CAViaR* (i.e., we use equation (14) as regression quantile objective function), 8) *CAViaR & EVT*, 9) *Efficient CAViaR & EVT*.

To optimise the likelihood of EVT, GARCH and CAViaR models, we used the command *fminsearch* in Matlab, which is an optimisation algorithm based on the simplex search method by Nelder and Meade.¹⁶

We chose a window of 300 observations for the historical simulation method. On the basis of the results of the EVT Monte Carlo study, we decided to estimate the GPD parameters using 120 exceptions, a number that allows us to reduce the MSE at very low quantiles and still keep bias and MSE of higher quantiles within a reasonable range. The CAViaR model augmented with EVT was estimated with a confidence level of 7.5%. Finally, the bandwidth to compute the height of the density function for the estimation of efficient CAViaR models was set equal to 0.1.

The results are shown in Tables 1 and 2. Let's start looking at the quantile results for the first 5 GARCH processes. Not surprisingly, the worst models in terms of both MSE and bias are Historical Simulation, Normal GARCH and EVT, which are the only not correctly specified models for the simulated processes. The only exception is when the underlying process is a Normal GARCH, in which case the Normal GARCH model is correctly specified and is the best performer at all confidence levels.

At extreme confidence levels, such as 0.05%, also the QML GARCH and the plain CAViaR become unreliable. This is due to the fact that when the confidence level is very low, only very few observations exceed the quantile (if any) and CAViaR estimates are imprecise for the same reason that empirical quantile estimates are.

Another interesting feature, although still not surprising, is that by using the efficient CAViaR estimation there is actually a sensible gain in the efficiency of the estimates and a reduction in the bias. Note that for the first 5 DGPs, the assumption that allows the implementation of the efficient CAViaR estimation (i.i.d. standardised quantile residuals) is true.

The best models overall are:

• the normal GARCH in the case of a Normal GARCH DGP,

¹⁶ The initial values were set equal to the true GARCH parameters for GARCH and CAViaR models and equal to [0.15, 0.5] for the EVT. For the efficient CAViaR, the initial values were set equal to the optimizing parameters of the inefficient CAViaR. The termination tolerance was 0.0001 for both the function value and the parameters, the maximum number of evaluations allowed was 1000 and the maximum number of iterations was 500.

 the efficient CAViaR augmented with EVT for GARCH DGP generated using the Student-t with 3 and 4 degrees of freedom,

• the QML GARCH augmented with EVT for the gammas with reversed sign GARCH.

Note that the performance of QML GARCH and QML GARCH augmented with EVT is very similar at 5% and 1% confidence levels. This is consistent with the findings discussed in section 4.1, where we showed that 5% and 1% empirical quantile and EVT estimates are very close for a reasonable choice of the parameter k.

The results are less clear for the non i.i.d. GARCH. Note that in this case, none of the models is correctly specified. At 5% the historical simulation method seems to have the lowest bias, but the CAViaR models have the lowest MSE. At 1% the CAViaR augmented with EVT is the best performer, while at 0.05% none of the models seems to do a good job, although historical simulation and normal GARCH produce the lowest bias and MSE.

For the CAViaR DGP, the best estimates are delivered, not surprisingly, by the CAViaR models. Under this DGP, only the plain CAViaR is correctly specified and also the assumption behind efficient estimation is violated. Despite this, all the CAViaR models produce similar estimates of 5% quantiles. At 1%, instead, the plain CAViaR is clearly superior to the CAViaR augmented with EVT. Finally, at 0.05% the performance of the plain CAViaR is considerably worse, again because of scarcity of data that far out in the tail. The CAViaR model augmented with EVT seems to do a good job at lowering the MSE of the estimate, although plain EVT delivers an estimate with similar MSE and historical simulation produces estimates with much lower bias.

It is interesting to compare the relative efficiency of these estimators, as measured by the ratio of the Mean Square Error (MSE). For 1% quantile estimates, the normal GARCH has a MSE ratio of 0.37 compared to the QML GARCH augmented with EVT, when the true DGP is a normal GARCH. However this ratio quickly deteriorates to 1.13 under a Student-t with 3 degrees of freedom, 2.04 under a Student-t with 4 degrees of freedom.

The MSE ratio of QML GARCH & EVT to efficient CAViaR & EVT is 0.75 under the normal, 4.12 under Student-t with 3 degrees of freedom, 0.45 under the gamma (2,2), 1.08 under the non i.i.d. GARCH and 2.65 under the CAViaR GDP.

Similar results hold for other quantiles. These results show that CAViaR is generally a more robust method for estimating quantiles, when returns have a fat tailed distribution.

Turning our attention to the Expected Shortfall tables, we see analogous results: CAViaR models produce better estimates for heavy-tailed processes and QML GARCH & EVT is better with normal and gamma distributions with an infinite left endpoint. The results are once again mixed for the last two processes. All the models for the non i.i.d. GARCH are seriously biased and with high MSE, these results being worse at lower confidence levels. Also in the case of CAViaR DGP, all the estimates are biased and with high MSE, although in this case the size of the distortion seems to be considerably lower, especially at very low confidence levels.

A totally new result is the estimation of the Expected Shortfall using the regression technique described in section 3. First of all notice that at 0.05% the regression estimates are not available in most of the cases, because there are no observations beyond the quantile and hence it is not possible to implement the method. At 5% and 1%, instead, the method works reasonably well. The bias of the regression estimate is almost always lower than the bias of the EVT estimate. This feature is particularly striking for the CAViaR models augmented with EVT, in which case the EVT method seems to produce a considerable bias relative to the regression method. The results on MSE are similar: EVT yields a lower MSE when applied to QML GARCH, while the regression method works better when applied to CAViaR models.

5. Conclusion

We discussed the most commonly used methodologies for Value at Risk estimation, paying attention to their underlying assumptions and their weaknesses. The performance of the different models was evaluated through a Monte Carlo experiment. The results show that CAViaR models produce the best estimates with heavy-tailed DGP.

We also introduced a regression-based method for estimating the expected shortfall. This method can be used as an alternative to the extreme value theory approach. The Monte Carlo results indicate that the regression method tends to outperform the EVT approach at very common confidence levels, such as 1% or 5%.

References

Artzner, P., F. Delbaen, J. Eber and D. Heath (1997), Thinking Coherently, RISK 10(11): 68-71.

Artzner, P., F. Delbaen, J. Eber and D. Heath (1999), Coherent Measures of Risk, Universite' de Strasbourg, preprint.

Basle Committee on Banking Supervision (1996), Amendment to the Capital Accord to Incorporate Market Risks.

Beder, T.S. (1995), VaR: Seductive but Dangerous, Financial Analyst Journal, Sep-Oct, 12-24.

Bollerslev, T. (1986), Generalized Autoregressive Conditional Heteroskedasticity, *Journal of Econometrics* **31**: 307-327.

Bollerslev, T., R.F. Engle and D. Nelson (1994), ARCH Models, Ch. 49 in R.F. Engle and D.L. McFadden eds., *Handbook of Econometrics*, iv Elsevier.

Bollerslev, T. and J.M. Woolridge (1992), Quasi-Maximum Likelihood Estimation and Inference in Dynamic Models with Time Varying Covariances, *Econometric Reviews* **11**: 143-172.

Boudoukh, J., M. Richardson and R. Whitelaw (1998), The Best of Both Worlds, RISK 11: 64-67.

Buchinsky, M. (1998), Recent Advances in Quantile Regression Models: a Practical Guideline for Empirical Research. *Journal of Human Resources* **33(1)**: 88-127.

Chernozhukov, V. and L. Umantsev (2000), Conditional Value-at-Risk: Aspects of Modeling and Estimation, Standford University, preprint.

Danielsson, J. and C. de Vries (1997), Tail Index and Quantile Estimation with Very High Frequency Data, *Journal of Empirical Finance* **4**: 241-257.

Danielsson, J. and C. de Vries (1998), Beyond the Sample: Extreme Quantile and Probability Estimation, London School of Economics, Discussion Paper 298.

Danielsson, J., L. de Haan, L. Peng and C. de Vries (1998), Using a Bootstrap Method to Choose the Sample Fraction in Tail Index Estimation, Erasmus University, preprint.

Dekkers, A.L.M., Einmahl, J.H.J. and L. de Haan (1989), A Moment Estimator for the Index of an Extreme Value Distribution, *Am. Statist.* 17, 1833-1855.

Diebold, F., T. Schuermann and J. Stroughair (1999), Pitfalls and Opportunities in the Use of Extreme Value Theory in Risk Management, in *Advances in Computational Finance*, Kluwer Academic Publishers, Amsterdam.

Embrechts, P., C. Kluppelberg and T. Mikosch (1997), *Modelling Extremal Events for Insurance and Finance*, Springer, Berlin.

Engle, R.F. (1982), Autoregressive Conditional Heteroscedasticity with Estimates of the Variance of United Kingdom Inflation, *Econometrica* **50**: 987-1007.

Engle, R.F. and G. Gonzalez-Rivera (1991), Semiparametric ARCH Models, *Journal of Business and Economic Statistics* **9**: 345-359.

Engle, R.F. and S. Manganelli (1999), CAViaR: Conditional Value at Risk by Quantile Regression, NBER, Working Paper 7341.

Fama, E. (1965), The Behavior of Stock Market Prices, Journal of Business 38, 34-105.

Fisher, R.A. and L.H.C. Tippett (1928), Limiting Forms of the Frequency Distribution of the Largest and Smallest Member of a Sample, *Proc. Camb. Phil. Soc.* 24: 180-190.

Gourieroux, C. and J. Jasak (1998), Truncated Maximum Likelihood, Goodness of Fit Tests and Tail Analysis, Unpublished Manuscript.

Hull, J. and A. White (1998), Incorporating Volatility Updating into the Historical Simulation Method for VaR, *The Journal of Risk* 1: 5-19.

Koenker, R. and G. Bassett (1978), Regression Quantiles, Econometrica 46: 33-50.

Leadbetter, M.R., G. Lindgren and H. Rootzen (1983), *Extremes and Related Properties of Random Sequences and Processes*, New York : Springer-Verlag.

McNeil, A.J. and R. Frey (2000), Estimation of Tail-Related Risk Measures for Heteroscedastic Financial Time Series: an Extreme Value Approach, *Journal of Empirical Finance*, **7**: 271-300.

Mandelbrot, B. (1963), The Variations of Certain Speculative Prices, Journal of Business 36: 394-419.

Newey, W. and J. Powell (1991), Efficient Estimation of Linear and Type I Censored Regression Models Under Conditional Quantile Restrictions, *Econometric Theory* **6**: 295-317.

Pickands, J. (1975), Statistical Inference Using Extreme Order Statistics, The Annals of Statistics 3: 119-131

RiskMetrics (1996), Technical Document, Morgan Guaranty Trust Company of New York.

White, H. (1994), Estimation, Inference and Specification Analysis, Cambridge University Press.

Appendix A: Tables

BIAS - 0.05% quantile	Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Historical Simulation	8.42	275.76	144.42	62.67	42.01	546.09	0.02
Normal GARCH	0	1004.6	470.36	342.78	173.11	621.3	4.12
QML GARCH	1.33	255.34	102.63	8.35	6.3	1220.48	3.28
QML GARCH & EVT	0.26	2.45	2.6	1.28	1.89	1319.13	0.73
EVT	0.3	15.44	2.6	2.14	0.75	1005.84	0.41
CAViaR	3.65	104.76	33.51	22.37	12.12	827.67	0.92
Efficient CAViaR	2.88	141.47	58.77	14.04	8.31	974.02	0.88
CAViaR & EVT	0.06	1.53	1.36	0.89	0.48	1468.89	0.4
Efficient CAViaR & EVT	0.07	1.7	1.51	0.69	0.66	1469.57	0.49
MSE - 0.05% quantile	Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Historical Simulation	25.16	1052.5	462.24	166.96	105.94	1086.08	2.1
Normal GARCH	0.45	1147.3	493.24	351.04	176.58	702.98	5.56
QML GARCH	9.24	3060.5	1939.8	67.01	46.09	2854.82	14.18
QML GARCH & EVT	4.3	497.92	160.96	25.87	27.73	1971.75	2.22
EVT	10.67	875.75	269.61	82.49	44.54	1973.77	1.15
CAViaR	19.07	1430.4	483.97	133.35	75.96	1546.32	7.46
Efficient CAViaR	18.93	1905.4	894.33	137.82	75.43	1991	7.01
CAViaR & EVT	4.51	274.93	109.87	47.63	27.09	2139.6	1.16
Efficient CAViaR & EVT	4.36	275.5	114.41	45.81	30.48	2163.96	1.43

 $\label{eq:table_$

BIAS - 1% quantile	Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Historical Simulation	0.4	8.13	3.84	1.64	1.07	32.02	0.14
Normal GARCH	0	6.14	6.37	56.01	30.08	32.33	1.6
QML GARCH	0	0.29	0.08	0	0	25.77	0.01
QML GARCH & EVT	0	0.14	0.01	0.01	0.09	25.5	0.14
EVT	0.06	0.73	0.31	0.08	0.05	26.11	0.13
CAViaR	0	0.1	0.05	0.24	0.09	25.66	0
Efficient CAViaR	0	0.09	0.04	0.04	0.01	25.66	0
CAViaR & EVT	0	0.03	0.01	0.04	0.02	25.65	0.1
Efficient CAViaR & EVT	0	0.02	0	0.01	0.03	25.65	0.1

MSE - 1% quantile	Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Historical Simulation	7.17	101.31	50.8	32.8	21.62	75.5	1.17
Normal GARCH	0.23	24.81	9.12	57.74	30.96	41.43	2.16
QML GARCH	0.6	23.45	5.13	2.98	1.85	31.91	0.37
QML GARCH & EVT	0.062	21.82	4.46	2.39	3.75	30.44	0.53
EVT	3.62	36.51	19.67	15.36	9.79	42.15	0.4
CAViaR	1.36	15.52	9.74	12.92	7.62	34.38	0.08
Efficient CAViaR	1.27	13.97	8.84	9.59	6.1	34.22	0.08
CAViaR & EVT	0.87	6.34	3.73	6.8	4.37	28.27	0.2
Efficient CAViaR & EVT	0.83	5.29	3.68	5.3	4.69	28.18	0.2

BIAS - 5% quantile	Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Historical Simulation	0.02	0.12	0.07	0.07	0.05	0.59	0.01
Normal GARCH	0	4.08	1.05	5.37	3.32	9.01	2.11
QML GARCH	0	0.02	0	0	0	0.68	0
QML GARCH & EVT	0	0.01	0	0	0	0.7	0
EVT	0.01	0.01	0	0.02	0.01	0.68	0
CAViaR	0	0	0	0	0	0.75	0
Efficient CAViaR	0	0	0	0	0	0.75	0
CAViaR & EVT	0	0	0	0.01	0	0.78	0
Efficient CAViaR & EVT	0	0	0	0.01	0	0.78	0
MSE - 5% quantile	Normal	t(3)	t(4)	Gamma(2.2)	Gamma (4,2)	non-iid GARCH	CAViaR
Historical Simulation			-(-)	Oummu(-,-)		non na ormeen	
	2.53	11.99	6.81	7.93	5.74	8.42	2
Normal GARCH	2.53 0.12	11.99	6.81 2.22	7.93 5.79	5.74 3.59	8.42 20.96	2 3.13
Normal GARCH QML GARCH	2.53 0.12 0.21	11.99 11.01 3.16	6.81 2.22 0.87	7.93 5.79 0.74	5.74 3.59 0.55	8.42 20.96 3.04	2 3.13 0.4
Normal GARCH QML GARCH QML GARCH & EVT	2.53 0.12 0.21 0.2	11.99 11.01 3.16 3.09	6.81 2.22 0.87 0.83	7.93 5.79 0.74 0.68	5.74 3.59 0.55 0.5	8.42 20.96 3.04 3.04	2 3.13 0.4 0.4
Normal GARCH QML GARCH QML GARCH & EVT EVT	2.53 0.12 0.21 0.2 1.7	11.99 11.01 3.16 3.09 9.45	6.81 2.22 0.87 0.83 5	7.93 5.79 0.74 0.68 5.17	5.74 3.59 0.55 0.5 3.67	8.42 20.96 3.04 3.04 6.2	2 3.13 0.4 0.4 1.66
Normal GARCH QML GARCH QML GARCH & EVT EVT CAViaR	2.53 0.12 0.21 0.2 1.7 0.44	11.99 11.01 3.16 3.09 9.45 1.35	6.81 2.22 0.87 0.83 5 0.91	7.93 5.79 0.74 0.68 5.17 1.55	5.74 3.59 0.55 0.5 3.67 1.13	8.42 20.96 3.04 3.04 6.2 1	2 3.13 0.4 0.4 1.66 0.1
Normal GARCH QML GARCH QML GARCH & EVT EVT CAViaR Efficient CAViaR	2.53 0.12 0.21 1.7 0.44 0.43	11.99 11.01 3.16 3.09 9.45 1.35 1.16	6.81 2.22 0.87 0.83 5 0.91 0.87	7.93 5.79 0.74 0.68 5.17 1.55 1.46	5.74 3.59 0.55 0.5 3.67 1.13 1.07	8.42 20.96 3.04 3.04 6.2 1 0.93	2 3.13 0.4 1.66 0.1 0.09
Normal GARCH QML GARCH QML GARCH & EVT EVT CAViaR Efficient CAViaR CAViaR & EVT	2.53 0.12 0.21 0.2 1.7 0.44 0.43 0.41	11.99 11.01 3.16 3.09 9.45 1.35 1.16 1.15	6.81 2.22 0.87 0.83 5 0.91 0.87 0.79	7.93 5.79 0.74 0.68 5.17 1.55 1.46 1.41	5.74 3.59 0.55 0.5 3.67 1.13 1.07 1.01	8.42 20.96 3.04 3.04 6.2 1 0.93 1.02	2 3.13 0.4 1.66 0.1 0.09 0.11

BIAS05% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Normal GAPCH	Normal	0	3375	1291	505	252	1879	12.71
Nomial OAKCH	Reg	-	2221	808	337	168	-	-
QML GARCH	Reg	-	-	-	-	-	-	-
OML GARCH & EVT	EVT	0.75	14.61	12.69	2.55	3.62	34774	10.69
QIVIL OARCH & LVI	Reg	-	-	-	-	-	-	-
FVT	EVT	8.57	2.61	20.78	23.07	22.37	18201	10.32
	Reg	-	-	-	-	-	-	-
CAViaR	Reg	-	-	-	-	-	-	-
Efficient CAViaR	Reg	-	-	-	-	-	-	-
CAViaR & EVT	EVT	32.26	102.93	77.21	260	154	4110	11.28
	Reg	-	-	-	-	-	-	-
Efficient CAViaR &	EVT	31.81	101	75.69	265	149	4191	11.62
EVT	Reg	-	-	-	-	-	-	-
MSE05% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
N 1CADOU	Normal	0.53	3712	1344	517	257	2028	14.47
Normal GARCH	Reg	-	2662	1092	349	174	-	-
QML GARCH	Reg	-	-	-	-	-	-	-
OML CARCH & EVT	EVT	7.49	2079	613	56.99	48.69	3328914	12.93
QIVIL GARCH & EVI	Reg	-	-	-	-	-	-	-
EVT	EVT	21.93	2831	749	140	87.17	1075793	11.43
EVI	Reg	-	-	-	-	-	-	-
CAViaR	Reg	-	-	-	-	-	-	-
Efficient CAViaR	Reg	-	-	-	-	-	-	-
CAVID & EVT	EVT	41.6	1306	435	392	222	7964	12.61
CAVIAN & EVI	Reg	-	-	-	-	-	-	-
Efficient CAViaR &	EVT	41	1317	445	393	225	8911	13.34
EVT	Reg	-	-	-	-	-	-	-
		1	L	L				
BIAS - 1% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Normal GARCH	Normal	0	113.84	63.79	122.98	64.02	107.37	11.81
	Reg	0	12.09	10.74	54.98	28.42	162.22	-
QML GARCH	Reg	0	0.47	0.04	0	0	123.42	10
OML GARCH & EVT	EVT	0.01	0.07	0.11	0.21	0.27	249.62	9.09
Quill OAICEI & EVI	Reg	0	0.28	0.01	0.05	0.11	114 54	14 07

 $\label{eq:Table 2-Bias} \textbf{Table 2}-\textbf{Bias} \text{ and } \textbf{MSE} \text{ of the Expected Shortfall Estimates}.$

BIAS - 1% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Normal GARCH	Normal	0	113.84	63.79	122.98	64.02	107.37	11.81
Nomilai OAKCH	Reg	0	12.09	10.74	54.98	28.42	162.22	-
QML GARCH	Reg	0	0.47	0.04	0	0	123.42	10
OML CARCH & EVT	EVT	0.01	0.07	0.11	0.21	0.27	249.62	9.09
QML UAKCH & EVI	Reg	0	0.28	0.01	0.05	0.11	114.54	14.07
EVT	EVT	8.13	21.1	22.07	33.12	23.94	242.38	6.27
17 1 1	Reg	0.21	2.86	1.13	0.54	0.36	125.7	13.4
CAViaR	Reg	0.03	0.64	0.21	0.23	0.08	116.99	8.47
Efficient CAViaR	Reg	0.03	0.41	0.24	0.02	0.02	117.28	8.47
CAVIOR & EVT	EVT	9.8	13.88	13.17	51.69	32.42	108.24	9.17
CAVIAK & EVI	Reg	0	0.02	0.01	0.04	0.01	113.17	14.04
Efficient CAViaR &	EVT	9.61	13.63	12.89	54.16	31.99	110.49	9.17
EVT	Reg	0	0.02	0.01	0.01	0.02	113.07	14.07

MSE - 1% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
Normal CAPCH	Normal	0.3	141.67	68.7	126.25	65.55	121.46	12.68
Notilial OAKCH	Reg	0.69	158.65	23.81	58.02	30.23	502.9	-
QML GARCH	Reg	0.91	180.05	22.56	5.67	3.62	175.76	11.89
OML GARCH & EVT	EVT	0.94	77.31	15.05	5.04	5.24	1566.38	10.17
QNIL OAKCII & LVI	Reg	1.07	180.09	23.56	6.02	6.05	163.83	17.16
EVT	EVT	12.77	118.42	63.1	56.49	38.35	1746.66	6.64
	Reg	5.39	133.87	56.26	27.82	17.14	184.85	15.45
CAViaR	Reg	1.9	50.31	25.15	20.83	12.4	153.33	9.67
Efficient CAViaR	Reg	1.8	60.54	26.88	15.59	9.92	157	9.69
CAViaD & EVT	EVT	12.17	49.59	29.68	73.7	45.04	288.97	9.57
CAVIAN & LVI	Reg	1.5	55.3	17.32	11.59	7.43	138.6	17.04
Efficient CAViaR &	EVT	11.91	45.43	28.73	72.69	45.23	239.49	9.54
EVT	Reg	1.47	51.68	16.51	10.11	7.63	138.32	16.91

BIAS - 5% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma(4,2)	non-iid GARCH	CAViaR
Normal GARCH	Normal	0	2.24	2.61	30.74	16.81	20.02	30.7
Normal OAICTI	Reg	0	9.49	1.72	5.22	2.98	63.01	56.05
QML GARCH	Reg	0	0.08	0.01	0	0	14.47	22.69
OML CARCH & EVT	EVT	0	0.02	0	0.01	0	16.83	23.77
QNIL OARCH & EVI	Reg	0	0.04	0	0.01	0	14.42	22.12
FVT	EVT	9.16	29.88	24.67	33.99	24.97	20.47	9.56
	Reg	0.01	0.27	0.08	0.03	0.03	15.07	24.88
CAViaR	Reg	0	0.02	0.02	0.02	0.01	14.33	21.35
Efficient CAViaR	Reg	0	0.02	0.02	0.02	0.01	14.33	21.41
CAVIOR & EVT	EVT	0.47	0.38	0.4	2.18	1.5	15.41	20.67
CAVIAN & EVI	Reg	0	0.01	0.01	0.01	0.01	14.33	21.19
Efficient CAViaR &	EVT	0.45	0.35	0.38	2.14	1.47	15.25	20.58
EVT	Reg	0	0.01	0.01	0.02	0.01	14.33	21.15
MSE - 5% quantile		Normal	t(3)	t(4)	Gamma(2,2)	Gamma (4,2)	non-iid GARCH	CAViaR
MSE - 5% quantile	Normal	Normal 0.19	t(3) 11.34	t(4) 4.35	Gamma(2,2) 31.86	Gamma (4,2) 17.41	non-iid GARCH 38.27	CAViaR 35.56
MSE - 5% quantile Normal GARCH	Normal Reg	Normal 0.19 0.29	t(3) 11.34 59.51	t(4) 4.35 6.74	Gamma(2,2) 31.86 6.48	Gamma (4,2) 17.41 3.79	non-iid GARCH 38.27 200.98	CAViaR 35.56 67.09
MSE - 5% quantile Normal GARCH QML GARCH	Normal Reg Reg	Normal 0.19 0.29 0.33	t(3) 11.34 59.51 15.19	t(4) 4.35 6.74 2.94	Gamma(2,2) 31.86 6.48 1.58	Gamma (4,2) 17.41 3.79 1.06	non-iid GARCH 38.27 200.98 28.7	CAViaR 35.56 67.09 26.99
MSE - 5% quantile Normal GARCH QML GARCH	Normal Reg Reg EVT	Normal 0.19 0.29 0.33 0.31	t(3) 11.34 59.51 15.19 9.71	t(4) 4.35 6.74 2.94 2.49	Gamma(2,2) 31.86 6.48 1.58 1.46	Gamma (4,2) 17.41 3.79 1.06 0.98	non-iid GARCH 38.27 200.98 28.7 40.05	CAViaR 35.56 67.09 26.99 27.84
MSE - 5% quantile Normal GARCH QML GARCH QML GARCH & EVT	Normal Reg Reg EVT Reg	Normal 0.19 0.29 0.33 0.31 0.35	t(3) 11.34 59.51 15.19 9.71 15.17	t(4) 4.35 6.74 2.94 2.49 2.93	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08	non-iid GARCH 38.27 200.98 28.7 40.05 28.07	CAViaR 35.56 67.09 26.99 27.84 26.33
MSE - 5% quantile Normal GARCH QML GARCH QML GARCH & EVT	Normal Reg EVT Reg EVT	Normal 0.19 0.29 0.33 0.31 0.35 11.73	t(3) 11.34 59.51 15.19 9.71 15.17 55.16	t(4) 4.35 6.74 2.94 2.49 2.93 35.66	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87
MSE - 5% quantile Normal GARCH QML GARCH & EVT EVT	Normal Reg EVT Reg EVT Reg	Normal 0.19 0.29 0.33 0.31 0.35 11.73 2.74	t(3) 11.34 59.51 15.19 9.71 15.17 55.16 28.96	t(4) 4.35 6.74 2.94 2.93 35.66 12.2	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98 10.89	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56 7.22	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74 29.17	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87 25.94
MSE - 5% quantile Normal GARCH QML GARCH & EVT EVT CAViaR	Normal Reg Reg EVT Reg EVT Reg Reg	Normal 0.19 0.29 0.33 0.31 0.35 11.73 2.74 0.68	t(3) 11.34 59.51 15.19 9.71 15.17 55.16 28.96 8.71	$\begin{array}{r} t(4) \\ 4.35 \\ 6.74 \\ 2.94 \\ 2.49 \\ 2.93 \\ 35.66 \\ 12.2 \\ 2.63 \end{array}$	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98 10.89 3.12	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56 7.22 2.1	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74 29.17 16.94	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87 25.94 24.23
MSE - 5% quantile Normal GARCH QML GARCH & EVT EVT CAViaR Efficient CAViaR	Normal Reg Reg EVT Reg EVT Reg Reg Reg	Normal 0.19 0.29 0.33 0.31 0.35 11.73 2.74 0.68 0.66	t(3) 11.34 59.51 15.19 9.71 15.17 55.16 28.96 8.71 10.88	$\begin{array}{r} t(4) \\ 4.35 \\ 6.74 \\ 2.94 \\ 2.49 \\ 2.93 \\ 35.66 \\ 12.2 \\ 2.63 \\ 2.51 \end{array}$	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98 10.89 3.12 2.97	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56 7.22 2.1 1.98	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74 29.17 16.94 17.13	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87 25.94 24.23 24.35
MSE - 5% quantile Normal GARCH QML GARCH & EVT EVT CAViaR Efficient CAViaR	Normal Reg EVT Reg EVT Reg Reg Reg EVT	Normal 0.19 0.29 0.33 0.31 0.35 11.73 2.74 0.68 0.66 1.27	t(3) 11.34 59.51 15.19 9.71 15.17 55.16 28.96 8.71 10.88 4.89	$\begin{array}{r} t(4) \\ 4.35 \\ 6.74 \\ 2.94 \\ 2.49 \\ 2.93 \\ 35.66 \\ 12.2 \\ 2.63 \\ 2.51 \\ 2.88 \end{array}$	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98 10.89 3.12 2.97 6.01	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56 7.22 2.1 1.98 3.91	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74 29.17 16.94 17.13 268.12	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87 25.94 24.23 24.25 23.47
MSE - 5% quantile Normal GARCH QML GARCH & EVT EVT CAViaR Efficient CAViaR CAViaR & EVT	Normal Reg EVT Reg EVT Reg Reg Reg EVT Reg	Normal 0.19 0.29 0.33 0.31 0.35 11.73 2.74 0.68 0.66 1.27 0.68	t(3) 11.34 59.51 15.19 9.71 15.17 55.16 28.96 8.71 10.88 4.89 10.44	$\begin{array}{r} t(4) \\ 4.35 \\ 6.74 \\ 2.94 \\ 2.49 \\ 2.93 \\ 35.66 \\ 12.2 \\ 2.63 \\ 2.51 \\ 2.88 \\ 2.5 \end{array}$	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98 10.89 3.12 2.97 6.01 3.03	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56 7.22 2.1 1.98 3.91 1.96	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74 29.17 16.94 17.13 268.12 17.1	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87 25.94 24.23 24.35 23.47 24.11
MSE - 5% quantile Normal GARCH QML GARCH & EVT EVT CAViaR Efficient CAViaR CAViaR & EVT Efficient CAViaR &	Normal Reg EVT Reg EVT Reg Reg Reg EVT Reg EVT	Normal 0.19 0.29 0.33 0.31 0.35 11.73 2.74 0.68 0.66 1.27 0.68 1.21	t(3) 11.34 59.51 15.19 9.71 15.17 55.16 28.96 8.71 10.88 4.89 10.44 4.44	$\begin{array}{r} t(4) \\ 4.35 \\ 6.74 \\ 2.94 \\ 2.49 \\ 2.93 \\ 35.66 \\ 12.2 \\ 2.63 \\ 2.51 \\ 2.88 \\ 2.5 \\ 2.78 \end{array}$	Gamma(2,2) 31.86 6.48 1.58 1.46 1.65 43.98 10.89 3.12 2.97 6.01 3.03 5.88	Gamma (4,2) 17.41 3.79 1.06 0.98 1.08 31.56 7.22 2.1 1.98 3.91 1.96 3.88	non-iid GARCH 38.27 200.98 28.7 40.05 28.07 34.74 29.17 16.94 17.13 268.12 17.1 235.13	CAViaR 35.56 67.09 26.99 27.84 26.33 9.87 25.94 24.23 24.23 24.35 23.47 24.11 23.28

Appendix B: Figures





European Central Bank Working Paper Series

- I "A global hazard index for the world foreign exchange markets" by V. Brousseau and F. Scacciavillani, May 1999.
- 2 "What does the single monetary policy do? A SVAR benchmark for the European Central Bank" by C. Monticelli and O. Tristani, May 1999.
- 3 "Fiscal policy effectiveness and neutrality results in a non-Ricardian world" by C. Detken, May 1999.
- 4 "From the ERM to the euro: new evidence on economic and policy convergence among EU countries" by I. Angeloni and L. Dedola, May 1999.
- 5 "Core inflation: a review of some conceptual issues" by M. Wynne, May 1999.
- 6 "The demand for M3 in the euro area" by G. Coenen and J.-L. Vega, September 1999.
- 7 "A cross-country comparison of market structures in European banking" by O. de Bandt and E. P. Davis, September 1999.
- 8 "Inflation zone targeting" by A. Orphanides and V. Wieland, October 1999.
- 9 "Asymptotic confidence bands for the estimated autocovariance and autocorrelation functions of vector autoregressive models" by G. Coenen, January 2000.
- 10 "On the effectiveness of sterilized foreign exchange intervention" by R. Fatum, February 2000.
- II "Is the yield curve a useful information variable for the Eurosystem?" by J. M. Berk and P. van Bergeijk, February 2000.
- 12 "Indicator variables for optimal policy" by L. E. O. Svensson and M. Woodford, February 2000.
- 13 "Monetary policy with uncertain parameters" by U. Söderström, February 2000.
- 14 "Assessing nominal income rules for monetary policy with model and data uncertainty" by G. D. Rudebusch, February 2000.
- 15 "The quest for prosperity without inflation" by A. Orphanides, March 2000.
- 16 "Estimating the implied distribution of the future short term interest rate using the Longstaff-Schwartz model" by P. Hördahl, March 2000.
- 17 "Alternative measures of the NAIRU in the euro area: estimates and assessment" by S. Fabiani and R. Mestre, March 2000.
- 18 "House prices and the macroeconomy in Europe: Results from a structural VAR analysis" by M. Iacoviello, April 2000.

- 19 "The euro and international capital markets" by C. Detken and P. Hartmann, April 2000.
- 20 "Convergence of fiscal policies in the euro area" by O. De Bandt and F. P. Mongelli, May 2000.
- 21 "Firm size and monetary policy transmission: evidence from German business survey data" by M. Ehrmann, May 2000.
- 22 "Regulating access to international large value payment systems" by C. Holthausen and T. Rønde, June 2000.
- 23 "Escaping Nash inflation" by In-Koo Cho and T. J. Sargent, June 2000.
- 24 "What horizon for price stability" by F. Smets, July 2000.
- 25 "Caution and conservatism in the making of monetary policy" by P. Schellekens, July 2000.
- 26 "Which kind of transparency? On the need for clarity in monetary policy-making" by B. Winkler, August 2000.
- 27 "This is what the US leading indicators lead" by M. Camacho and G. Perez-Quiros, August 2000.
- 28 "Learning, uncertainty and central bank activism in an economy with strategic interactions" by M. Ellison and N. Valla, August 2000.
- 29 "The sources of unemployment fluctuations: an empirical application to the Italian case" by S. Fabiani, A. Locarno, G. Oneto and P. Sestito, September 2000.
- 30 "A small estimated euro area model with rational expectations and nominal rigidities" by G. Coenen and V. Wieland, September 2000.
- 31 "The disappearing tax base: Is foreign direct investment eroding corporate income taxes?" by R. Gropp and K. Kostial, September 2000.
- 32 "Can indeterminacy explain the short-run non-neutrality of money?" by F. De Fiore, September 2000.
- 33 "The information content of M3 for future inflation" by C. Trecroci and J. L. Vega, October 2000.
- 34 "Capital market development, corporate governance and the credibility of exchange rate pegs" by O. Castrén and T. Takalo, October 2000.
- 35 "Systemic risk: A survey" by O. De Bandt and P. Hartmann, November 2000.
- 36 "Measuring core inflation in the euro area" by C. Morana, November 2000.
- 37 "Business fixed investment: Evidence of a financial accelerator in Europe" by P. Vermeulen, November 2000.

- 38 "The optimal inflation tax when taxes are costly to collect" by F. De Fiore, November 2000.
- 39 "A money demand system for euro area M3" by C. Brand and N. Cassola, November 2000.
- 40 "Financial structure and the interest rate channel of ECB monetary policy" by B. Mojon, November 2000.
- 41 "Why adopt transparency? The publication of central bank forecasts" by P. M. Geraats, January 2001.
- 42 "An area-wide model (AWM) for the euro area" by G. Fagan, J. Henry and R. Mestre, January 2001.
- 43 "Sources of economic renewal: from the traditional firm to the knowledge firm" by D. R. Palenzuela, February 2001.
- 44 "The supply and demand for eurosystem deposits The first 18 months" by U. Bindseil and F. Seitz, February 2001.
- 45 "Testing the Rank of the Hankel matrix: a statistical approach" by G. Camba-Mendez and G. Kapetanios, February 2001.
- 46 "A two-factor model of the German term structure of interest rates" by N. Cassola and J. B. Luís, February 2001.
- 47 "Deposit insurance and moral hazard: does the counterfactual matter?" by R. Gropp and J. Vesala, February 2001.
- 48 "Financial market integration in Europe: on the effects of EMU on stock markets" by M. Fratzscher, March 2001.
- 49 "Business cycle and monetary policy analysis in a structural sticky-price model of the euro area" by M. Casares, March 2001.
- 50 "Employment and productivity growth in service and manufacturing sectors in France, Germany and the US" by T. von Wachter, March 2001.
- 51 "The functional form of the demand for euro area MI" by L. Stracca, March 2001.
- 52 "Are the effects of monetary policy in the euro area greater in recessions than in booms?" by G. Peersman and F. Smets, March 2001.
- 53 "An evaluation of some measures of core inflation for the euro area" by J.-L. Vega and M. A. Wynne, April 2001.
- 54 "Assessment criteria for output gap estimates" by G. Camba-Méndez and D. R. Palenzuela, April 2001.
- 55 "Modelling the demand for loans to the private sector in the euro area" by A. Calza, G. Gartner and J. Sousa, April 2001.

- 56 "Stabilization policy in a two country model and the role of financial frictions" by E. Faia, April 2001.
- 57 "Model-based indicators of labour market rigidity" by S. Fabiani and D. Rodriguez-Palenzuela, April 2001.
- 58 "Business cycle asymmetries in stock returns: evidence from higher order moments and conditional densities" by G. Perez-Quiros and A. Timmermann, April 2001.
- 59 "Uncertain potential output: implications for monetary policy" by M. Ehrmann and F. Smets, April 2001.
- 60 "A multi-country trend indicator for euro area inflation: computation and properties" by E. Angelini, J. Henry and R. Mestre, April 2001.
- 61 "Diffusion index-based inflation forecasts for the euro area" by E. Angelini, J. Henry and R. Mestre, April 2001.
- 62 "Spectral based methods to identify common trends and common cycles" by G. C. Mendez and G. Kapetanios, April 2001.
- 63 "Does money lead inflation in the euro area?" by S. N. Altimari, May 2001.
- 64 "Exchange rate volatility and euro area imports" by R. Anderton and F. Skudelny, May 2001.
- 65 "A system approach for measuring the euro area NAIRU" by S. Fabiani and R. Mestre, May 2001.
- 66 "Can short-term foreign exchange volatility be predicted by the Global Hazard Index?" by V. Brousseau and F. Scacciavillani, June 2001.
- 67 "The daily market for funds in Europe: Has something changed with the EMU?" by G. P. Quiros and H. R. Mendizabal, June 2001.
- 68 "The performance of forecast-based monetary policy rules under model uncertainty" by A. Levin, V. Wieland and J. C.Williams, July 2001.
- 69 "The ECB monetary policy strategy and the money market" by V. Gaspar, G. Perez-Quiros and J. Sicilia, July 2001.
- 70 "Central Bank forecasts of liquidity factors: Quality, publication and the control of the overnight rate" by U. Bindseil, July 2001.
- 71 "Asset market linkages in crisis periods" by P. Hartmann, S. Straetmans and C. G. de Vries, July 2001.
- 72 "Bank concentration and retail interest rates" by S. Corvoisier and R. Gropp, July 2001.
- "Interbank lending and monetary policy transmission evidence for Germany" by M. Ehrmann and A. Worms, July 2001.

- 74 "Interbank market integration under asymmetric information" by X. Freixas and C. Holthausen, August 2001.
- 75 "Value at risk models in finance" by S. Manganelli and R. F. Engle, August 2001